Author’s response to reviews

Title: Adverse childhood events and risk of diabetes onset in the 1979 National Longitudinal Survey of Youth Cohort

Authors:

E. Anne Lown (anne.lown@ucsf.edu)
Camillia Lui (CLui@arg.org)
Kate Karriker-Jaffe (kkarrikerjaffe@arg.org)
Nina Mulia (NMulia@arg.org)
Edwina Williams (EWilliams@arg.org)
Yu Ye (YYe@arg.org)
Libo Li (LLi@arg.org)
Thomas Greenfield (TGreenfield@arg.org)
William Kerr (WKerr@arg.org)

Version: 1 Date: 18 Oct 2018

Author’s response to reviews:

We addressed each of your suggestions and the manuscripts should follow all the formatting rules.